Author's response to reviews

Title: Successful outcome of a pregnancy in advanced cirrhosis due to Hepatitis B surface antigenemia, delta super-infection and Hepatitis C co-infection:a case report

Authors:

Dr AMNA Subhan (amna.subhan@aku.edu)
Dr SHAHAB Abid (shahab.abid@aku.edu)
Dr WASIM Jafri (wasim.jafri@aku.edu)

Version: 2 Date: 4 July 2007

Author's response to reviews:

manuscript reviewed. title restructured, page numbers and keywords section removed.